Massachusetts Institute of Technology - Physics Department Physics - 8.02 Information on Exam #2 April 1, 2002. When & Where The second Exam will be held on Wednesday, April 10 during regular 8.02 lecture hours. For students whose last name begins with

A - K in 26-100 L - Z in Walker (3rd floor) Walker Memorial is located on East Campus. Please enter from the river side (Memorial Drive).

Please take the exam at your assigned lecture hour.

Material Covered

This exam will cover:

(i) all material discussed in lectures through April 1 (ii) all associated reading (iii) homework assignments 4, 5, & 6.

• It is a closed book exam.

• We will provide you with many equations.

• You may not use a calculator, and you are not allowed to bring any notes.

There are no make-up exams.

A missed exam counts as a zero. Only in case of verifiable illness can you be excused by Professor Wen (room 12-106, x3-5016, wen@dao.mit.edu) from taking an exam. If at all possible, this should be done before the exam.

Grading Policy See the General Information, handed out during the first lecture. You can find it on the 8.02 Home Page under ”general information”. Exam Review I will give an exam review during regular 8.02 lecture hours on Monday, April 8. It can be viewed on MIT Cable TV (channel 9) every hour on the hour, starting April 8 at 5 PM. Tutoring Schedule See the 8.02 Home Page under ”tutoring”.

Good Luck! Walter Lewin